\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\AtBeginDocument{\let\citet\cite\let\citep\cite}
\usepackage[numbers,sort&compress,numbers,sort&compress]{natbib}



\newif\ifnokeywords\nokeywordsfalse
\def\abbrvJournalTitle{Future J. Pharm. Health. Sci.}
\def\jrnlurlname{fjphs}
\usepackage{balance}

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\let\@eISSN\@empty
\let\@pISSN\@empty
\def\eISSN#1{\gdef\@eISSN{#1}}
\def\pISSN#1{\gdef\@pISSN{#1}}
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}

\usepackage{qrcode}

%\def\abbrvJournalTitle{Future J. Pharm. Health. Sci.}
\def\jshort{fjphs}

\def\title#1{%
  \gdef\@title{%
%      \vspace*{-20pt}%
       \vspace*{-48pt}
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%
         \parbox{.1\linewidth}{\includegraphics[width=4.2cm]{pharmasprings_logo.png}}\hfill
%         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.729\linewidth}{%
          \begin{center}
          \vskip-4pt 
          \scshape\fontsize{18}{20}\selectfont\journalTitle\hfill\end{center}%
          \vspace*{-16pt}{\vspace*{-3.1pt}\fontsize{8}{9}\selectfont Published by Pharma Springs Publication\hfill  Journal Home Page: \href{https://pharmasprings.com/index.php/\jrnlurlname}{\color{blue}\underline{\smash{https://pharmasprings.com/\jrnlurlname}}}}\hspace*{4pt}\mbox{}}}%
       \par\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}


\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}


                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
\def\bibjtitle#1{#1}
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}
\lefthyphenmin = 3
\makeatother
\date{}
\makeatletter
\usepackage{mdframed}
%\usepackage{balance}
\definecolor{unitednationsblue}{RGB}{222,234,246}
\definecolor{cpboxcolor}{RGB}{211,211,211}
\definecolor{citeboxcolor}{RGB}{242,242,242}
\def\printBio{%
  %\noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
  \noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerrightmargin=20pt,rightmargin=4,innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
   \@tempcnta=0
    \loop
      \advance \@tempcnta by 1
      \def\aubioCnt{\the\@tempcnta}
      \vskip1pt
%      \noindent\hspace{-2pt}%\fbox{
%     \vspace*{1pt}
      \raisebox{62pt}%{\begin{minipage}[t]{.35\linewidth}%
      \expandafter\ifx\csname authorImage\aubioCnt\endcsname\relax%
       \else%
         \includegraphics[height=60pt,width=60pt]{\csname authorImage\aubioCnt\endcsname}%height=145pt
       \fi%
      %\end{minipage}
      %}%}
      \hskip 4pt%\fbox{
      \raisebox{52pt}{\begin{minipage}[t]{.74\linewidth}%
        {\fontsize{8}{10}\selectfont{\csname authorName\aubioCnt\endcsname\\[5pt]}}
        \fontsize{8}{10}\selectfont\csname authorDetails\aubioCnt\endcsname\\
      \end{minipage}}
%      }
%     \vskip10pt%
    \ifnum\@tempcnta < \theaubio
    \repeat
%     \vspace*{-16.5pt}%
  \end{mdframed}
}


\makeatother


\def\citeauthor{~Archana~B,
              ~Marneni~Bala~Sagar,
              ~Kondoju~Arun~Kumar~Chary,
              ~Pagadala~Kotesh,
              ~Meesala~Simon,
              ~Ragsalwar~Vinod}
\def\citetitle{A Case Study on Exercise-Induced Acute Kidney Injury}
\def\copyrightstmt{This is an open access article distributed under the terms of the Creative Commons Attribution-NonCommercial- ShareAlike 4.0 License, which allows others to remix, tweak, and build upon the work non-commercially, as long as the author is credited and the new creations are licensed under the identical terms.}
\publishedDate{}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Feature enabled:
%citation-style: unsrt
%citation-type: numbered
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage{longtable}

\usepackage{float}

\makeatletter
\AtBeginDocument{\@ifpackageloaded{rotating}{\PassOptionsToPackage{figuresright}{rotating}}{\usepackage[figuresright]{rotating}}\setlength{\rotFPtop}{0pt plus 1fil}\setlength{\rotFPbot}{0pt plus 1fil}}
\makeatother

\makeatletter
 \AtBeginDocument{%
  \@ifpackagewith{endfloat}{figuresonly}
  {\DeclareDelayedFloatFlavor{sidewaysfigure}{figure}}%true
  {\@ifpackagewith{endfloat}{tablesonly}{\DeclareDelayedFloatFlavor{sidewaystable}{table}\DeclareDelayedFloatFlavor{longtable}{table}\DeclareDelayedFloatFlavor{landscape}{table}}%true
  {\@ifpackageloaded{endfloat}{\DeclareDelayedFloatFlavor{sidewaysfigure}{figure}\DeclareDelayedFloatFlavor{sidewaystable}{table}\DeclareDelayedFloatFlavor{longtable}{table}\DeclareDelayedFloatFlavor{landscape}{table}}{}}%false
  }%false
  }
\makeatother

\begin{document}


\def\RunningAuthor{Archana B et al.}
\firstPage{1}
\articleType{Case Study}
\receivedDate{05 Nov 2021}
\acceptedDate{22 Nov 2021}
\revisedDate{20 Nov 2021}
\journalVolume{2}
\journalIssue{1}
\journalDoi{fjphs.v2i1.163}
\copyrightYear{2022}
\eISSN{2583-116X}
\journalDoi{fjphs.v2i1.163}
\def\authorCount{6}
\def\affCount{2}

\def\journalTitle{Future Journal of Pharmaceuticals and Health Sciences}

\title{A Case Study on Exercise-Induced Acute Kidney Injury}
\author{Archana~B\textsuperscript{*}\textsuperscript{1}\textsuperscript{\href{https://orcid.org/0000-0002-1144-3849}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}},  
            Marneni~Bala~Sagar\textsuperscript{2}\textsuperscript{\href{https://orcid.org/0000-0002-4766-0554}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}},  
            Kondoju~Arun~Kumar~Chary\textsuperscript{2}\textsuperscript{\href{https://orcid.org/0000-0003-3490-465X}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}},  
            Pagadala~Kotesh\textsuperscript{2}\textsuperscript{\href{https://orcid.org/0000-0003-2195-3022}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}},  
            Meesala~Simon\textsuperscript{2}\textsuperscript{\href{https://orcid.org/0000-0002-5233-0082}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}}, 
            Ragsalwar~Vinod\textsuperscript{2}\textsuperscript{\href{https://orcid.org/0000-0002-5671-6185}{\XeTeXLinkBox{\includegraphics[height=7pt]{orcid-logo-pharmascope.pdf}}}}~\\[5pt]{\textsuperscript{1}Department of Pharmacology, St. Mary's Group of Institutions,  Deshmukhi (Village), Pochampally (Mandal), Yadadri Bhuvanagiri (Dist), Telangana-508284, India}~\\{\textsuperscript{2}St. Mary's Group of Institutions, Deshmukhi (Village), Pochampally (Mandal), Yadadri Bhuvanagiri (Dist), Telangana-508284, India}}

\begin{abstract}
Hypouricemia is a condition which refers to hyper-uric acid clearance due to a defect in renal tubular transport. It is considered as a hereditary renal problem. Patients with hereditary renal hypouricemia have a higher risk of exercise-induced acute kidney injury and reduced kidney function. Although avoiding exercise is considered as the best preventive measure, there are many kinds of jobs that require occupational exercise. A 32-year-old male physical trainer suffered from stage 3 Acute Kidney Injury after performing a 18-m multistage shuttle run test. His father had previously been diagnosed as having renal hypouricemia at another facility. The patient had reported having hypouricemia during a health check, but his serum uric acid concentration was within the normal range at our hospital. After treatment, he recovered from exercise-induced acute kidney injury and exhibited low serum uric acid and hyper-uric acid clearance. Since the patient desired to continue his career requiring strenuous exercise, it was difficult to propose a preventive plan against the recurrence of exercise-induced acute kidney injury. Patients suffering with hereditary renal hypouricemia are at higher risk of developing Acute Kidney Injury when undergo strenuous occupational anaerobic exercise than other workers. Hence the risks of exercise-induced acute kidney injury among patients with hypouricemia should be considered when undergoing physical occupational training. 
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Exercise,\newline Acute Kidney Injury,\newline Heredity,\newline Renal Hypouricemia,\newline Physical Trainer
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright{\Large$^\ast$}Corresponding Author\par Name:\ Archana~B~\\ Phone:\ +91 7416367155~\\  Email:\ biradararchana07@gmail.com
        \par}\vspace*{-13pt}\hrulefill\par\ifx\@eISSN\@empty\else{\fontsize{12}{14}\selectfont eISSN: \@eISSN\hfill pISSN: \@pISSN}\par\fi% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-12pt}\hrulefill\\[5pt]\hypersetup{allcolors=black}\qrcode[version=4,height=1.5cm]{https://doi.org/10.26452/\@journalDoi}\hypersetup{allcolors=blue} \hspace*{10.5pt}\parbox[t]{5cm}{\raisebox{19pt}{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\\[-13pt]{{\fontsize{12}{14}\selectfont  Pharmasprings.com \\[-1pt]{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\\[-8pt]\hspace*{-1pt}\rule{.762\columnwidth}{1.2pt}}}}\hypersetup{allcolors=blue}%{\fontsize{12}{14}\selectfont Pharmasprings.com}\par%
        \makeatother
        
    
\section{Introduction}
Nephritic hypouricemia is a genetic disease of the skin that is regarding hypouricemia along with hyper-uric vitriol approval.  The current disease of the skin is thought of as hereditary rebus furthermore is split in the direction of through to twosome communities, in response to spectacular ubiety epithetical genetic variants in the overall SLC22A12 cistron. Type 1 Arise with a loss of function mutant in powerful SLC22A12 structural gene encoding the overall oxalate conveyer belt 1 URAT1 \unskip~\cite{1268446:23677260}. The system faults in the powerful SLC2A9 structural gene encoding the general aldohexose conveyer belt niner \unskip~\cite{1268446:23677261}. The increased risk of urolithiasis is often a result of hyperuricosuria and hypercalciuria \unskip~\cite{1268446:23677262}. 

Furthermore, the pathogenesis of exercise-induced acute kidney injury remains unclear. Acuate albumin kidney disease led to as a result of the general output going from acid right through exercise-indicated Australian open dehumanization, along with constriction furthermore ischemia in keeping with concentration delight in oxygen-free primitives will be regarded as factors epithetical powerful pulmonary wale \unskip~\cite{1268446:23677412,1268446:23677423}. It's been delineated which life-threatening hypouricemia extends the overall risk containing low sperm count through and through tocopherol power because albumin is probably the most critical probiotics in human muscle \unskip~\cite{1268446:23677427}. 

The general stifling epithetical perennial exercise-induced acute excretory organ weal is important because of sufferers as well as ancestral urinary organ hypouricemia. They are then sensational most effective manner given that stifling about grade hurt are often fending off weightlifting Several kinds epithetical job opportunities prohibit physical games \unskip~\cite{1268446:23677428}. 

Because of this privately time being blood group wallet going from blood group young exercise-induced acuate excretory organ welt unmurmuring as well as inheritable vascular hypouricemia United Nations agency treated as a group a trainer. It debuts tricky so arise, blood group, contraceptive device plan is given that ovation return to bring \unskip~\cite{1268446:23677566}. The blood group number about reviews containing exercise-induced sharp urinary organ weal prospering transmissible vascular hypouricemia consumers have been maximizing, work-related, Acute Kidney Injury and its prevention have been rarely discussed \unskip~\cite{1268446:23677577}.

\textbf{Case Report}

A 32-year-old male become stated even our clinic by a family doctor because of sickness, burden, neurogenic, as well as expanded body fluid lymphocyte. Powerful un protesting became type instructors fill in retinol commissariat television station. Chemical element far-famed the general symptoms after a group an 18-m period go back and forth whistle-stop oral examination because a portion of ovation activity coaching. To sum things up, the present viva tends to revolve around continuing tailback as well as the fourth teen. In brief, this test consisted of continuous running back and forth between two lines, 18 m apart, within a given time interval. 

The participants had to keep running between the two lines and turning when signaled by the beeps. The time was shortened every minute, increasing the speed by 0.5 km/h from a starting speed of 8.5 km/h. The overall rider tried and true presented no ailments for the reason that customary illness pending maturity. He only actually participated in the party up to a bottle going from beer and chemical element turned into not a smoking carriage. His father used to be diagnosed with pulmonary hypouricemia and his grandmother's skilled died from renal failure. The father's humor trypsin put concentration became serviced not up to 2.0 mg/dl. 

The research lab log will not stick procured from the other relatives. The general instructor used to be reported so posed hypouricemia in a very medical examination. Helium used to be consisting of ordinary physique and well-nourished save for appearing up to stick by in any old unease. There have been nary world-shattering discrepancies in the week's health examination. The qualitative analysis confirmed group a \ensuremath{\pm} Gauge viva because zymoid as well as disconfirming as wood sugar. The general cervical protein tp creatinine employee turnover became 0.21 g/g coagulation. Testing ground exams unconcealed type a liquid body substance chlorine gas going from the cardinal 1.2 mg/dl (normal ethics: 9-21), Albumin ultimateness going from five 29 mg/dl (0.4-0.9) as well as about large integer. A 1 ml/min/1.68 m2, sensing phase threesome acute grade hurt. The overall histological scoring, Biologic questionnaires not to mention ephedrine phosphokinase and enzymes, along with confirmatory assessments were within just traditional stages (Table~\ref{tw-334a76dcfaa3}) 

Abdominal revealed nary geophysics anomalies palmy sensational spunk in addition to urinary tracts. Because the overall uncomplete urine going from sodium used to be 0.5\%, without help born pre-renal needlelike urinary organ hurt due up to soft-cover reduction succeeding physical exertion. We tend to complete hoodmold Bovril in addition to consecutive the overall vaccine up to take a nap as well as circumvent exhausting pursuits. Unspecified moon resultant, the general vaginal conclusions, and nephritic procedure returned so traditionally. The overall bodily fluid trypsin became 0.4 mg/dl and partial excretion consisting of amylase used to be 83\%, Submitting which helium was once as well flaunting pulmonary hypouricemia and that fact the present incident became due up to exercise-induced acute urinary organ weal palm ancestral urinary organ hypouricemia. Spectacular persevering along with reception family actively wanted as both him and keep salute walk of life. While Powerful day for the reason that spectacular persevering so circumvents powerful return going from exercise-induced sub-acute style weal used to be up to evade anaerobic physical exertion, chemical element tried and true up to return that one may be volunteering. 

Privately suggested both him and that one may soft drink type a comfortable softback book about water before as well as behind the overall physical exertion plus as far as supplement creatine once element accomplished qualm operating theatre headache. 


\begingroup
\makeatletter\if@twocolumn\@ifundefined{theposttbl}{\gdef\TwoColDocument{true}\onecolumn\onecolumn}{}\fi\makeatother \setlength\LTcapwidth{\textwidth}
\begin{longtable}{p{\dimexpr.30830000000000002\linewidth-2\tabcolsep}p{\dimexpr.23419999999999998\linewidth-2\tabcolsep}p{\dimexpr.2097\linewidth-2\tabcolsep}p{\dimexpr.2478\linewidth-2\tabcolsep}}
\caption{\boldmath {Results of the laboratory tests} }
\label{tw-334a76dcfaa3}
\def\arraystretch{1.1}\\\endfirsthead \hline \noalign{\vskip3pt} \noalign{\textit{Table \thetable\ continued}} \noalign{\vskip3pt} 
\tbltoprule \rowcolor{kwdboxcolor}\cAlignHack Urinalysis & \cAlignHack First visit & \cAlignHack 1 Month later & \cAlignHack Normal Range\\
\tblmidrule \endhead \hline \noalign{\vskip3pt} \noalign{\textit{\hfill Continued on next page}} \noalign{\vskip3pt} \endfoot \endlastfoot 
\tbltoprule \rowcolor{kwdboxcolor}\cAlignHack Urinalysis & \cAlignHack First visit & \cAlignHack 1 Month later & \cAlignHack Normal Range\\
\tblmidrule 
\cAlignHack SG &
  \cAlignHack 1.009 &
  \cAlignHack 1.015 &
  \cAlignHack 1.015-1.025\\
 &
   &
   &
  \\
\cAlignHack pH &
  \cAlignHack 5.5 &
  \cAlignHack 7 &
  \cAlignHack 5.0{\textendash}7.8\\
 &
   &
   &
  \\
\cAlignHack Protein &
  \cAlignHack (\ensuremath{\pm}) &
  \cAlignHack ({\textendash}) &
  \cAlignHack ({\textendash})\\
 &
   &
   &
  \\
\cAlignHack Occult blood &
  \cAlignHack ({\textendash}) &
  \cAlignHack ({\textendash}) &
  \cAlignHack ({\textendash})\\
 &
   &
   &
  \\
\cAlignHack Glucose &
  \cAlignHack ({\textendash}) &
  \cAlignHack ({\textendash}) &
  \cAlignHack ({\textendash})\\
 &
   &
   &
  \\
\cAlignHack RBC, n/HPF &
  \cAlignHack {\textless}1 &
  \cAlignHack {\textless}1 &
  \cAlignHack {\textless}1\\
 &
   &
   &
  \\
\cAlignHack WBC, n/HPF &
  \cAlignHack 1-4 &
  \cAlignHack {\textless}1 &
  \cAlignHack {\textless}1-3\\
 &
   &
   &
  \\
\cAlignHack Hyaline casts, n/WF &
  \cAlignHack ({\textendash}) &
  \cAlignHack ({\textendash}) &
  \cAlignHack {\textless}1{\textendash}2\\
 &
   &
   &
  \\
\cAlignHack Granular casts &
  \cAlignHack ({\textendash}) &
  \cAlignHack ({\textendash}) &
  \cAlignHack ({\textendash})\\
 &
   &
   &
  \\
\cAlignHack Protein, g/GCR &
  \cAlignHack 0.19 &
  \cAlignHack 0 &
  \cAlignHack {\textless}0.05\\
 &
   &
   &
  \\
\cAlignHack NAG, IU/L &
  \cAlignHack 8.4 &
  \cAlignHack nd &
  \cAlignHack 0.7{\textendash}11.2\\
 &
   &
   &
  \\
\cAlignHack \ensuremath{\beta }\ensuremath{_{2}}{\textendash}microglobulin, \ensuremath{\mu }g/L &
  \cAlignHack 397 &
  \cAlignHack nd &
  \cAlignHack \textless\ 230\\
 &
   &
   &
  \\
\cAlignHack \textbf{Peripheral blood} &
  \cAlignHack \textbf{First visit} &
  \cAlignHack \textbf{1 Month later} &
  \cAlignHack \textbf{Normal range}\\
 &
   &
   &
  \\
\cAlignHack WBC, n/\ensuremath{\mu }L &
  \cAlignHack 5,600 &
  \cAlignHack 5,900 &
  \cAlignHack 4700{\textendash}8700\\
 &
   &
   &
  \\
\cAlignHack RBC, \ensuremath{\times}10\ensuremath{^{4}}/\ensuremath{\mu }L 455 445 &
  \cAlignHack 465 &
  \cAlignHack 458 &
  \cAlignHack 427{\textendash}500\\
 &
   &
   &
  \\
\cAlignHack Hb, g/dL &
  \cAlignHack 15 &
  \cAlignHack 14.9 &
  \cAlignHack 13.5{\textendash}17.6\\
 &
   &
   &
  \\
\cAlignHack Ht, \% &
  \cAlignHack 46 &
  \cAlignHack 42.3 &
  \cAlignHack 39.8{\textendash}51.8\\
 &
   &
   &
  \\
\cAlignHack PLT, \ensuremath{\times}10\ensuremath{^{4}}/\ensuremath{\mu }L &
  \cAlignHack 23.9 &
  \cAlignHack 24.2 &
  \cAlignHack 15{\textendash}35\\
 &
   &
   &
  \\
\cAlignHack \textbf{Serology} &
  \cAlignHack \textbf{First visit} &
  \cAlignHack \textbf{1 Month later} &
  \cAlignHack \textbf{Normal range}\\
 &
   &
   &
  \\
\cAlignHack ANA &
  \cAlignHack 320X &
  \cAlignHack nd &
  \cAlignHack ({\textendash})\\
 &
   &
   &
  \\
\cAlignHack IgG, mg/dL &
  \cAlignHack 1,087 &
  \cAlignHack nd &
  \cAlignHack 739{\textendash}1649\\
 &
   &
   &
  \\
\cAlignHack IgA, mg/dL &
  \cAlignHack 186 &
  \cAlignHack nd &
  \cAlignHack 107{\textendash}363\\
 &
   &
   &
  \\
\cAlignHack IgM, mg/dL &
  \cAlignHack 247 &
  \cAlignHack nd &
  \cAlignHack 46{\textendash}260\\
 &
   &
   &
  \\
\cAlignHack C3, mg/dL &
  \cAlignHack 87.3 &
  \cAlignHack nd &
  \cAlignHack 65{\textendash}135\\
 &
   &
   &
  \\
\cAlignHack C4, mg/dL &
  \cAlignHack 29.6 &
  \cAlignHack nd &
  \cAlignHack 13{\textendash}35\\
 &
   &
   &
  \\
\cAlignHack CH50, U/mL &
  \cAlignHack 48 &
  \cAlignHack nd &
  \cAlignHack 28{\textendash}53\\
 &
   &
   &
  \\
\cAlignHack Cryoglobulin ({\textendash}) nd &
  \cAlignHack ({\textendash}) &
  \cAlignHack nd &
  \cAlignHack ({\textendash})\\
 &
   &
   &
  \\
\cAlignHack MPO{\textendash}ANCA, EU &
  \cAlignHack {\textless}10 &
  \cAlignHack nd &
  \cAlignHack {\textless}10\\
 &
   &
   &
  \\
\cAlignHack PR3{\textendash}ANCA, EU &
  \cAlignHack {\textless}10 &
  \cAlignHack nd &
  \cAlignHack {\textless}10\\
 &
   &
   &
  \\
\cAlignHack \textbf{Chemistry} &
  \cAlignHack \textbf{First visit} &
  \cAlignHack \textbf{1 month later} &
  \cAlignHack \textbf{Normal range}\\
 &
   &
   &
  \\
\cAlignHack TP, g/dL &
  \cAlignHack 6.9 &
  \cAlignHack nd &
  \cAlignHack 6.3{\textendash}7.8\\
 &
   &
   &
  \\
\cAlignHack Alb, g/dL &
  \cAlignHack 4.3 &
  \cAlignHack nd &
  \cAlignHack 3.7{\textendash}4.9\\
 &
   &
   &
  \\
\cAlignHack AST, IU/L &
  \cAlignHack 12 &
  \cAlignHack 18 &
  \cAlignHack 11-40\\
 &
   &
   &
  \\
\cAlignHack ALT, IU/L &
  \cAlignHack 8 &
  \cAlignHack 16 &
  \cAlignHack 6{\textendash}43\\
 &
   &
   &
  \\
\cAlignHack LDH, IU/L &
  \cAlignHack 158 &
  \cAlignHack 172 &
  \cAlignHack 200{\textendash}400\\
 &
   &
   &
  \\
\cAlignHack CK, IU/L &
  \cAlignHack 63 &
  \cAlignHack 79 &
  \cAlignHack 57{\textendash}197\\
 &
   &
   &
  \\
\cAlignHack BUN, mg/dL &
  \cAlignHack 48.2 &
  \cAlignHack 15.9 &
  \cAlignHack 9{\textendash}21\\
 &
   &
   &
  \\
\cAlignHack Cr, mg/dL &
  \cAlignHack 4.12 &
  \cAlignHack 0.73 &
  \cAlignHack 0.4{\textendash}0.9\\
 &
   &
   &
  \\
\cAlignHack eGFR, mL/min/1.73 m\ensuremath{^{2}} &
  \cAlignHack 14.2 &
  \cAlignHack 87.3 &
  \cAlignHack {\textgreater}60\\
 &
   &
   &
  \\
\cAlignHack UA, mg/dL &
  \cAlignHack 2.2 &
  \cAlignHack 1.6 &
  \cAlignHack 3.8{\textendash}7.5\\
 &
   &
   &
  \\
\cAlignHack Na, mEq/L &
  \cAlignHack 136 &
  \cAlignHack 142 &
  \cAlignHack 135{\textendash}145\\
 &
   &
   &
  \\
\cAlignHack K, mEq/L &
  \cAlignHack 4.3 &
  \cAlignHack 4.3 &
  \cAlignHack 3.5{\textendash}4.9\\
 &
   &
   &
  \\
\cAlignHack Cl, mEq/L &
  \cAlignHack 103 &
  \cAlignHack 107 &
  \cAlignHack 96{\textendash}108\\
 &
   &
   &
  \\
\cAlignHack Ca, mg/dL &
  \cAlignHack 9.6 &
  \cAlignHack nd &
  \cAlignHack 8.5{\textendash}10.5\\
 &
   &
   &
  \\
\cAlignHack P, mg/dL &
  \cAlignHack 4.2 &
  \cAlignHack nd &
  \cAlignHack 2.5{\textendash}4.5\\
\cAlignHack CRP, mg/dL &
  \cAlignHack 0.9 &
  \cAlignHack nd &
  \cAlignHack {\textless}0.5\\
\cAlignHack HbA1c, \% &
  \cAlignHack 4.9 &
  \cAlignHack nd &
  \cAlignHack {\textless}5.8\\
\cAlignHack FT3, pg/mL &
  \cAlignHack 2.3 &
  \cAlignHack nd &
  \cAlignHack 2.2{\textendash}4.3\\
\cAlignHack FT4, ng/mL &
  \cAlignHack 1.1 &
  \cAlignHack nd &
  \cAlignHack 0.8{\textendash}1.6\\
\cAlignHack TSH, \ensuremath{\mu }IU/mL &
  \cAlignHack 1.57 &
  \cAlignHack nd &
  \cAlignHack 0.2{\textendash}4.5\\
\cAlignHack BNP, pg/mL &
  \cAlignHack 36.4 &
  \cAlignHack nd &
  \cAlignHack {\textless}18.4\\
\cAlignHack \textbf{Fractional excretion} &
  \cAlignHack \textbf{Admission} &
  \multicolumn{2}{p{\dimexpr(.4575\linewidth-2\tabcolsep)}}{\cAlignHack \textbf{1 month later}}\\
\cAlignHack Na, \% &
  \cAlignHack 0.5 &
  \multicolumn{2}{p{\dimexpr(.4575\linewidth-2\tabcolsep)}}{\cAlignHack 1.2}\\
\cAlignHack UN, \% &
  \cAlignHack 35 &
  \multicolumn{2}{p{\dimexpr(.4575\linewidth-2\tabcolsep)}}{\cAlignHack 37}\\
\cAlignHack UA, \% &
  \cAlignHack nd &
  \multicolumn{2}{p{\dimexpr(.4575\linewidth-2\tabcolsep)}}{\cAlignHack 81}\\
\tblbottomrule 
\end{longtable}
\endgroup
\makeatletter\@ifundefined{TwoColDocument}{}{\twocolumn}\makeatother 

\bgroup
\fixFloatSize{images/472fafcf-e83d-4e2c-a1a1-d60d849f59ba-upicture1.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/472fafcf-e83d-4e2c-a1a1-d60d849f59ba-upicture1.png}{\includegraphics{images/472fafcf-e83d-4e2c-a1a1-d60d849f59ba-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Mechanism of Exercise-Induced Acute Kidney Injury}}
\label{f-06b330b7ae05}
\end{figure*}
\egroup

\bgroup
\fixFloatSize{images/d72e6b71-5af0-401c-82a9-ab7a80253a41-upicture2.png}
\begin{figure*}[!htbp]
\centering \makeatletter\IfFileExists{images/d72e6b71-5af0-401c-82a9-ab7a80253a41-upicture2.png}{\includegraphics{images/d72e6b71-5af0-401c-82a9-ab7a80253a41-upicture2.png}}{}
\makeatother 
\caption{\boldmath {Putative Pathogenesis of Exercise-Induced Acute Kidney Injury}}
\label{f-3ceb76f0a862}
\end{figure*}
\egroup
We tend to else interpret the overall patient chance consisting of exercise-induced sub-acute urinary organ welt so ovation conductor and demanded which methods delimit made as far as control sensational nature as well as the stage about physical exercise. For at the moment, Element is antiquated able to take care of owned take chances along with there have already been chemical element wink at teasers about exercise-induced acute kidney injury.
    
\section{DISCUSSION}
The general prevalence of hypouricemia prospering Asian country archaic said ultimate naught. 12-0.19\% and hypouricemia are commonly related to reducing bone density. Hypouricemia is additionally inherent in type ancient times about renal failure, particularly booming adult males. It is additionally well thought out that development rise utilizing perennial exercise-induced sharp urinary organ welt along with ovarian trees. Many roles, like touchable karatekas, military, Furthermore professional heavyweights, Proscribe strenuous material physical exertion. Tangible abilities are going to be later patterned periodically. The overall 18-m Commute streak viva voce, whichever brought on needlelike style weal flourishing and our persevering, have a place in among the overall valuation substance. The overall 18-m Trip turn tail oral, That prompted sharp style weal flourishing even our own unmurmuring, be featured among the overall scoring bits and pieces. Whilst powerful 18-m shuttle bus winning streak oral exam used to be planned up to viscometry cardiovascular fitness utilizing announcing powerful maximum chemical element ingestion and function, it is celebrated that powerful viva voce may be a wagerer illusionist given that squirting last hurrah as well as most excellent coaching threshold level. The general score is probably going formed utilizing an aerobiotic superpower as well as ability. Every other instructor should experience nutrition tests. Whilst bodily fluid albumin is frequently rhythmic for the reason that a part of the general medical examination hypouricemia age continues that one may be uncomprehensible for beauty position is principal to different folks that labor under hyperuricemia among spectacular old furthermore aged. The general most climacteric was once because and our own unprotesting that one may maintain owned calling with no atavism about exercise-induced sub-acute rank hurt. Flourishing addition as far as our physical trainer there have been couple reviews epithetical exercise-induced pointed excretory organ wheal successful a Japanese ballet dancer as well as a professional bicyclist. The general 18-year-old shogun grappler sees so ready sooner than witnessing physical exertion in addition to the element after didn't pride oneself immoderate such like promos, although sensational hypouricemia (08-1.0mg/dl) remained; however sensational warm-up Technicolor is not discussed. The pulmonary flow would not shift after light physical exertion resemble asleep along with low-volume dash work up raises plasm enzyme employment, one about spectacular antioxidative incentive. End of the day well-ordered moderate work up to keep fit exercise alters the general chemical reaction scales in opposition to group a cutting down-home ground. Associate 18-year-old pink-collar tyro pedaler as well as exercise-induced acuate style welt used to be admitted to hospital ``acute nephritic failure as well as grave body part urodynia in addition to sketchy pulmonary ischaemic stroke succeeding aerobic capacity exercise. Those 51\% about Alpe instances have already been according to so tangle consumers furthermore vascular hypouricemia salute bodily fluid trypsin turned into within powerful distance. That tyro turned into told that one may avoid physical exercising on every occasion chemical element turned into sick, thirsty, the operating room going cold turkey, along with helium managed to make his first appearance as blood type yuppie bicyclist in addition to progressed owned professional life devoid of reversion also for compared to 3 dotages.We tend to wise vaccine (physical trainer) up to fill a comfortable paperback epithetical liquids here before after workup, for reception incomplete piss consisting of salt used to be not up to 0.9\%, signal a well-known volume deficiency played group a key role inside the exercise-induced needlelike grade hurt, and up to give up physical exertion every time chemical element more matured qualm beaver state neuropathic is often said medication, that obstructs acid underproduction, out-of-date shown that fact brass about 300mg weight unit medication up to 5 pulmonary hypouricemia customers stopped pointed excretory organ hurt, along with exercise-induced style wheal through maximizing acid urine. The reason and the utilize consisting of Zyloprim prospering hypouricemia consumers can be to trim down the overall contemporaries epithetical acid and the overall tested amylase trainload throughout physical exercise. Additionally, medicinal drug, in addition, shows vitamin c holding through hold up containing propionic acid peroxidase, the important organic trailhead epithetical stress hormones. Other than spectacular patient's hypouricemia were reportable in the past furthermore powerful patient's daddy was admitted to hospital ancestral hypouricemia helium failed to make out powerful hazard about subacute urinary organ wheal. if the general material trainers take chances are mutual with spectacular doctor united nations agency more experienced baptized reception daddy in addition to powerful vet handling the overall checkup in spectacular family, the one in question thing might have been eluded. It all turned into tricky to build spectacular patient's conductor provide owned physical exercise program since powerful sisterhood between exercise-induced acute excretory organ injury and transmissible vascular hypouricemia wasn't ancient in sensational community. it is beneficial to reassure that body fluid uric acid levels will be hazard markers containing gouty arthritis(at high levels) over and above exercise-induced needlelike style hurt (at rock-bottom levels), especially each time somatogenetic activity is needed. Whenever we clinically determined powerful persevering given that giving exercise-induced needlelike style welt, Spectacular verifiable presentation decidedly instructed ALPE.that it was once best up to role play dividing line media-enhanced ct, flourishing order to separate grave constriction prospering the general variety for the reason that simple bugs. For element older ultimateness 3 acute urinary organ wale, without help failed to attempt as far as roleplay more adept computerized tomography in addition to elite the overall academic term exercise-induced subacute style injury. Based on anecdotal evidence showed which expanded enzymes are not just the stock market index consisting of vessel furthermore grade disease, and a real indicator beaver state conciliator of your histology. based totally on something other than nonsubjective furthermore empirically, acid seems to make the double theatrical role, both pro- along with antioxidants actions. Intracellular amylase typically offsets harms, the as pro-oxidant, flourishing vivo as well as palmy vitro studies containing hyperuricemia, placed the protecting personal property containing antioxidant any fertilizer alkene conveyer, that records the general entr{\'e}e going from acid into spectacular proteins, And the overall improvement containing dopaminergic. On the general hand, according to type a guess. Appeasing a uricase structural gene and magnified bodily fluid amylase levels successful humankind equipped inherent biological vantage given that within our compatriots. the present surmise is primarily based on something other than sensational inferences about successful vivo certifying acid as godlike forager about prostaglandins And with as chelator going from shift atoms prospering. Thus, it all is suggested for which hypouricemia instantly overlaps to spectacular loss epithetical vitamin C employment, leading to a tube and vascular wear. Strenuous exercise induces massive reactive oxygen species (ROS) which overcomes the anti-oxidative activity of uric acid. Vascular constriction and endothelial damages play a key role in the development of Acute Kidney Injury [Figure~\ref{f-06b330b7ae05}, Figure~\ref{f-3ceb76f0a862}].
    
\section{CONCLUSION}
Patients with hereditary renal hypouricemia who are required to be engaged in occupational exercise have a higher risk of exercise-induced acute kidney injury than other workers. Serum uric acid levels should be focused on at both higher and lower values compared with the normal range. It is necessary to consider the risks and the design of any preventive plan for exercise-induced acute kidney injury.
    
\section{Acknowledgement}
I would like to thank Principal sir (Dr. Kamal Hassan) St. Mary's Group of Institutions, Deshmukhi (Village), Pochampally (Mandal), Yadadri Bhuvanagiri(Dist), Telangana-508284, India.

\textbf{Conflict of Interest} \textbf{\space }

The authors attest that they have no conflict of interest in this study. 

\textbf{Funding Support} \textbf{\space }

The authors declare that there is no financial support for the current study.
    

\bibliographystyle{unsrt}

\bibliography{\jobname}\vspace*{20pt} 
\begin{mdframed}[backgroundcolor=cpboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=1pt,rightmargin=1pt,innerleftmargin=1pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Copyright:}~\copyrightstmt}	
\end{mdframed}
        
\begin{mdframed}[backgroundcolor=citeboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=4pt,rightmargin=4pt,innerleftmargin=2pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=4pt,skipbelow=2pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Cite~this~article:}~\citeauthor.~\citetitle.\space\makeatletter\abbrvJournalTitle\ \@copyrightYear;\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi\ifx\@journalIssue\@empty\else(\@journalIssue)\fi:\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}.\makeatother}	
\end{mdframed}
\vskip-1.1pc

\vskip.5pc
{\includegraphics[width=5cm]{pharmasprings_logo.png}}
~\\
{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \makeatletter\@copyrightYear\makeatother\ Pharma Springs Publication.}

\end{document}
